Author's response to reviews

Title: Paramedic assessment of pain in the cognitively impaired adult patient

Authors:

Bill A Lord (bill.lord@med.monash.edu.au)

Version: 3 Date: 5 October 2009

Author's response to reviews:

The following changes have been made as requested:

1. Capitalisation of the country on the title page has been corrected, and keywords removed.

2. Reference number 39 (Kunz 2009) has been included in the body of the text (p. 15).

3. Text indicating placement of the figure has been removed from the body of the manuscript.

4. Reference number 32 has been updated to show the volume number.

5. Minor changes to syntax have been made in the body of the manuscript and typographical errors corrected.

Thank you for accepting this revised proof.